Author's response to reviews

Title: Improved metabolic health among the obese in six population surveys 1986 to 2009. The Northern Sweden MONICA study.

Authors:

Martin Benckert (martin.benckert@gmail.com)
Mikael Lilja (mikael.lilja@jll.se)
Stefan Söderberg (stefan.soderberg@medicin.umu.se)
Mats Eliasson (mats.eliasson@nll.se)

Version: 5 Date: 22 January 2015

Author's response to reviews:

The Editor wrote:
One last mistake to fix in the discussion is the sentence:

"Notably we have not found any previous study reporting tre over long time long-term trends in metabolically health in a well-defined population using a standardized methodology As recently reviewed...?"

Should be revised to:

"Notably we have not found any previous study reporting tre over long time long-term trends in metabolically healthy in a well-defined population using a standardized methodology. As recently reviewed..."

I was not sure what was expected of me but added the reference for the last statement just after ....reviewed [24]... And pulled the two sentences together in the same paragraph at the bottom of page 6.

Best wishes

Mats Eliasson